Author's response to reviews

Title: Phosphatidylinositol 3'-kinase, mTOR, and Glycogen synthase kinase-3 beta mediated regulation of p21 in human urothelial carcinoma cells

Authors:

Nicole L Yohn (nyohn@mail.med.upenn.edu)  
Caitlyn N Bingaman (bingam_c@denison.edu)  
Ashley L DuMont (ashley.dumont@nyumc.org)  
Lina I Yoo (yool@denison.edu)

Version: 3 Date: 13 August 2011

Author's response to reviews: see over
August 12, 2011

BioMed Central
Editorial Office
Manuscript #4748855185276711

Dear Editor,

We were very excited to hear that our manuscript has been accepted for publication. We thank you and the reviewers for your rapid decision. We will now address the formatting changes that you required:

**Major Revisions**

1) Title: please ensure that the title is identical in the manuscript and the submission system

*We have changed the title in the online system to match that of the manuscript.*

2) Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution

*All authors were affiliated with the same department and university at the time they did the work. Therefore, one address is listed. We have added the department information and street address to the title page of the manuscript.*

3) Abstract: Please ensure that the abstract is identical in the manuscript file on the online submission system

*We have changed the abstract in the online system to match that in the revised manuscript.*

4) Authors’ contributions: Please indicate that all authors read and approved the final manuscript.

*This line has been added to the Authors’ contributions section. All authors have in fact read and approved the final manuscript.*

**Minor Revisions**

5) Corresponding author: Please link to the author list using a symbol

*An asterisk has been used instead of a number on the title page of the manuscript.*

6) Email address: please include email addresses for all the authors on the title page, using the following format, they do not need to be linked to the author list:

AB: abcd@institution.ac.uk
We have done this on the title page.

7) Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image

We have cropped the figures as suggested.

8) Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

We have checked the manuscript for typographical errors.

We thank you again for the opportunity to have our work published in BMC Urology. We hope we have properly addressed all of the formatting issues.

Sincerely,

Lina Yoo, PhD
Assistant Professor of Biology

PO Box 810
Denison University
Granville, OH 43023, USA
Phone: (+1) (740)587-4287
Email: yool@denison.edu